Author's response to reviews

Title: Quantitative Structure-Activity Relationship of Molecules Constituent of Different Essential Oils with Antituberculosis Activity against Mycobacterium tuberculosis and Mycobacterium bovis.

Authors:

Sergio Andrade-Ochoa (s_andrade_rat@hotmail.com)
Guadalupe V Nevarez-Moorillón (vnevare@uach.mx)
Luvia E. Sánchez-Torres (luviasanchez@hotmail.com)
Manuel Villanueva-Garcia (villagmster@gmail.com)
Blanca E. Sánchez-Ramírez (bsanche@uach.mx)
Luz M Rodríguez-Valdez (lmrodrig@uach.mx)
Blanca E Rivera-Chavira (bchaviraster@gmail.com)

Version: 2
Date: 16 April 2015

Author's response to reviews: see over
Dear Editor:

Attach you will find the paper entitled “Quantitative Structure-Activity Relationship of Molecules Constituent of Different Essential Oils with Antituberculosis Activity against Mycobacterium tuberculosis and Mycobacterium bovis” for your evaluation and possible publication in the Journal BMC Complementary and Alternative Medicine.

The article describes the antituberculosis activity of terpenes and phenylpropanes, major components of essential oils, as well as their molecular and structural characteristics obtained by computational chemistry. Results of biological activity were then related to their molecular properties to obtain QSAR models, that can be used for the intelligent design of new antituberculosis drugs. The document is not under evaluation by any other journal, and all the authors have reviewed and approved the manuscript. We have also included the information required by the Editorial Office.

We hope you consider our results as timely innovative and that the manuscript can be published in the BMC Complementary and Alternative Medicine.

Hoping to hear from you at your earliest convenience, I remain,

Sincerely

Blanca Estela Rivera-Chavira, Ph.D.
Professor
Universidad Autónoma de Chihuahua